Author’s response to reviews

Title: Does Anterior Chamber-Associated Immune Deviation (ACAID) play a role in posterior lamellar keratoplasty? Case Report of a splenectomized patient

Authors:

Deniz Hos (deniz.hos@uk-koeln.de)
Friederike Schaub (friederike.schaub@uk-koeln.de)
Claus Cursiefen (claus.cursiefen@uk-koeln.de)

Version: 2 Date: 22 Apr 2019

Author’s response to reviews:

1. Please could you send a copy of the consent for publication you obtained from the patient in this report to the BMC Ophthalmology Editorial Office. This is so it can be assessed by the Editor.

We have sent a scanned copy of the consent for publication obtained from the patient to the BMC Ophthalmology Editorial Office per email.

2. We note that you have not included an acknowledgements section. If you have no acknowledgements please put ‘Not Applicable’ in this section.

We have included an acknowledgements section and put "Not applicable" in this section (Page 7, from line 17)

3. Please combine the two Consent for publication sections so that they are one section.

We have combined the two consent for publication sections to one (Page 7, from line 1)

4. Please remove the point-by-point, cover letter and CARE checklist as these are no longer needed at this time.

We removed the point-by-point letter, the cover letter and the CARE checklist
5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

We have uploaded a single, final, clean version of the manuscript.